Author's response to reviews

Title: Effects of Mothers' Socio-Economic Status on the Management of febrile conditions in their under five children in a Resource Limited Setting.

Authors:

Adenike E Olaogun (nikeoolaogun@yahoo.com)
Abayomi A Adebayo (bayoade@oauife.edu.ng)
Olufemi E Ayandiran (olufeman@yahoo.com)
Olayinka A Olasode (olayinkaolasode@yahoo.com)

Version: 13 Date: 5 January 2006

Author's response to reviews:

The section on Competing interest has been included and the references have been revised in line with the journal style.